\begin{thebibliography}{10}
\addcontentsline{toc}{chapter}{Bibliography}
\bibitem{Sinerva}
P. Sinerva, these proceedings;\\
CDF Collaboration (F. Abe et al.), \prl{74}{1995}{2626};\\
D0 Collaboration (S. Abachi et al.), \prl{74}{1995}{2632}.

\bibitem{bardin}
D. Bardin et al., in
                  Reports of the Working Group on Precision
                  Calculations for the $Z$-Resonance, CERN 95-03,
                  pg. 7.

\bibitem{HollikBrus}
W. Hollik, talk at Int. Europhysics Conf. on HEP, Brussels, 1995,
KA-TP-15-1995, hep-ph/9512232. 


\bibitem{Swartz}
M. Swartz , these proceedings, and references therein.
\bibitem{Kniehl}
For a recent compilation of theoretical results see B. Kniehl, in
                  Reports of the Working Group on Precision
                  Calculations for the $Z$-Resonance, CERN 95-03,
                  pg. 299.
\bibitem{Sirlin}
A. Sirlin, ibid. pg. 299.
\bibitem{degrassi}
G. Degrassi et al., ibid. pg. 163.

\bibitem{Hill}
W.A. Bardeen, C.T. Hill, M. Lindner, \prd{41}{1990}{1647}.

%%%%%%%%% part 1 %%%%%%%%%%%%%%%%%
\bibitem{kaue1}  G.L. Kane and M.E. Peskin, \npb{195}{1982}{29}.
\bibitem{albaj3} C. Albajar et al. [UA1 Coll.],
 Phys. Lett. {\bf B256} (1991) 121, E: ibid. {\bf B262} (1991) 497.
\bibitem{PDG}
{Particle Data Group}, Phys. Rev. {\bf D50},  1171  (1994).


\bibitem{djoua1} A. Djouadi, J.H. K\"uhn and P.M. Zerwas,
               \zpc{46}{1990}{411}.

\bibitem{SuDong} Dong Su, private communication.

\bibitem{Trist} 
TOPAZ Coll., E. Nakano et al., Phys. Lett. {\bf B340} (1994) 135.\\
AMY Coll., F. Liu et al., Phys. Rev. {\bf D49} (1994) 4339.\\
 For a recent review see e.g. K. Miyabayashi, in {\em Recent
electroweak results from TRISTAN} in XXXth Rencontres de Moriond.

\bibitem{SchailZ} D.~Schaile and P.M.~Zerwas, %DESY Report 91--106.
       Phys. Rev. {\bf D45} (1992) 3262.
%%%%%%%%%% mass limits
\bibitem{chano1} M. Chanowitz, M. Furman and I. Hinchliffe,
                 \npb{153}{1979}{402}.
\bibitem{cabib1} N. Cabibbo, L. Maiani, G. Parisi and R. Petronzio,
                 \npb{158}{1979}{296}.
\bibitem{lindn1} M. Lindner, \zpc{31}{1986}{295}.

\bibitem{Munster94} G. M\"unster, 
       Nucl. Phys. B (Proc. Suppl.) {\bf 34} (1994) 596;
ibid., {\bf 42} (1995) 162.

\bibitem{HallThis} L. Hall, these proceedings.

\bibitem{albre1} H. Albrecht et al. (ARGUS), \plb{192}{1987}{245}.
            M. Artuso et al. (CLEO), \prl{62}{1989}{2233}.

\bibitem{veltm1} M. Veltman, \app{8}{1977}{475}.
\bibitem{veltm2} M. Veltman, \npb{123}{1977}{89}.


\bibitem{avdeev} L. Avdeev et al., \plb{336}{1994}{560};
                  Erratum ibid. {\bf B349} (1995) 597.

\bibitem{chetyr} K.G. Chetyrkin, J.H. K\"uhn, and M. Steinhauser,
                  \plb{351}{1995}{331}, \prl{75}{1995}{3394}.


\bibitem{acta} J.H.~K\"uhn, \app{12}{1981}{347}.\\
               J.H.~K\"uhn, \apa{\ Suppl.XXIV}{1982}{203}.


\bibitem{jk3}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \plb{207}{1988}{91}.

\bibitem{smith}
B.~H. Smith and M.~B. Voloshin, \plb{340}{1994}{176}.

\bibitem{ac95a}
A.~Czarnecki, \app{26}{1995}{845}.

\bibitem{BLM}
S.~J. Brodsky, G.~P. Lepage, and P.~B. Mackenzie, \prd{28}{1983}{228}.

\bibitem{den1}
A.~Denner and T.~Sack, \npb{358}{1991}{46}.

\bibitem{eil}
R.~Migneron G.~Eilam, R.R.~Mendel and A.~Soni, \prl{66}{1991}{3105}.

\bibitem{DennerHoang} 
A. Denner and A.H. Hoang, Nucl. Phys. {\bf B397} (1993) 483.

\bibitem{JezK}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \prd{48}{1993}{R1910};
\newblock E: ibid. {\bf D49} (1994) 4970.

\bibitem{Fujii}
K.~Fujii,
\newblock these proceedings.

\bibitem{stre}
{J.H.~K\"uhn} and K.H. Streng, \npb{198}{1982}{71}.

\bibitem{jk1}
M.~Je{\.z}abek and J.~H. K{\"u}hn, \npb{320}{1989}{20}.

\bibitem{cjk}
A.~Czarnecki, M.~Je{\.z}abek, and J.~H. K{\"u}hn, \app{20}{1989}{961},
E: ibid, {\bf B23} (1992) 173.\\
M.~Je{\.z}abek and C. J\"unger, \app{24}{1993}{1923}.

\bibitem{czajk91}
A.~Czarnecki, M.~Je{\.z}abek, and J.~H. K{\"u}hn, \npb{351}{1991}{70}.

\bibitem{cjkk94}
A.~Czarnecki, M.~Je{\.z}abek, J.~G. K{\"o}rner, and J.~H. K{\"u}hn,
\prl{73}{1994}{384}.

\bibitem{cj94}
A.~Czarnecki and M.~Je{\.z}abek, \npb{427}{1994}{3}.

\bibitem{ns30} G.L.~Kane, Proceedings, Madison Workshop 1979; \\
          I.~Bigi et al.\ in Ref.\cite{bigi2}; \\
          J.F.~Gunion, H.E.~Haber, G.~Kane and S.~Dawson, {\it The Higgs
          Hunter's Guide}, Addison--Wesley 1990; \\
          C.S.~Li and T.C.~Yuan, \prl{42}{1990}{3088}.

\bibitem{bigi2}  I. Bigi, Y.L. Dokshitzer, V.A. Khoze,
J. K\"uhn and P. Zerwas, \plb{181}{1986}{157}.

\bibitem{reya1}  E. Reya, P.M. Zerwas [conv.] et al., in Proceedings ECFA
              LHC Workshop, Aachen 1990, CERN 90--10.
\bibitem{PIK}P. Igo-Kemenes, J.H. K\"uhn [conv.], Proceedings,
                {\em \epem\ Collisions at 500~GeV: The  Physics Potential},
                DESY 92-123, pg.~255-326, pg.~327-378.

\bibitem{CzarDav}
A. Czarnecki and S. Davidson,
 Phys. Rev. {\bf D48} (1993) 4183; ibid., {\bf D47} (1993) 3063.\\
A. Czarnecki and A.I. Davydychev,
 \plb{325}{1994}{435}.

\bibitem{EWHiggs}
A. Czarnecki, \prd{48}{1993}{5250}.


\bibitem{ns38} J.~Gunion et al., Ref.\cite{ns30}; \\
          A.~Mendez and A.~Pomarol, \plb{252}{1990}{461}; \\
          C.S.~Li and R.J.~Oakes, \prd{43}{1991}{855}.

\bibitem{ns31} K.~Hikasa and M.~Kobayashi, \prd{36}{1987}{724}; \\
    H.~Baer, M.~Drees, R.~Godbole, J.F.~Gunion and X.~Tata, Tallahassee
    Report FSU--HEP--910308 (1991).
\bibitem{ns39} K.~Hidaka, Y.~Kizukuri and T.~Kon, Gakugei Report TGU--12
          (1991).

% hadron colliders 
\bibitem{ns40} J.~Ellis and S.~Rudaz, \plb{128}{1983}{248}; \\
          A.~Bouquet, J.~Kaplan and C.~Savoy, \npb{262}{1985}{299}.

\bibitem{ns41} G.~Eilam, J.L.~Hewett and A.~Soni, \prd{44}{1991}{1473};

\bibitem{ns32} H.~Fritzsch, \plb{224}{1989}{423}.

\bibitem{ns33} W.~Buchm\"uller and M.~Gronau, \plb{220}{1989}{641}.

\bibitem{gluec1} M. Gl\"uck, J.F. Owens and E. Reya,
                \prd{15}{1978}{2324};\\
                 B.L. Combridge, \npb{151}{1979}{429}.

\bibitem{wille1} S. Willenbrock and D.A. Dicus, \prd{34}{1986}{155};\\
                 C.--P. Yuan, \prd{41}{1990}{42}.

\bibitem{Martin95} A.D. Martin, W.J. Stirling, and R.G. Roberts,
hep-ph/9506423. 

\bibitem{Been91}
         W. Beenakker, W. L. van Neerven, R. Meng, G. A. Schuler and J. Smith,
                   Nucl. Phys. {\bf B351} (1991) 507.
\bibitem{nason1} P. Nason, S. Dawson and R. K. Ellis,
                   \npb{303}{1988}{607};
                 {\em ibid.} {\bf B}327 (1989) 49; Erratum B335 (1990)
                 260.\\
G. Altarelli, M. Diemoz, G. Martinelli, and P. Nason, Nucl. Phys. {\bf
  B308} (1988) 724.

\bibitem{beena1} W. Beenakker, H. Kuijf, W. L. van Neerven and J. Smith,
                    \prd{40}{1989}{54}.
\bibitem{meng1} R. Meng, G. A. Schuler, J. Smith and W. L. van Neerven,
              \npb{339}{1990}{325}.

\bibitem{diemo1} M. Diemoz, F. Ferrari, E. Longo and G. Martinelli,
                    \zpc{39}{1988}{21};\\
           M. Gl\"uck, E. Reya and A. Vogt, \zpc{48}{1990}{471} and
              Proceedings of the 'Small-x' Meeting, DESY, 1990.
\bibitem{Appel} D. Appel, G. Sterman, and P. Mackenzie,
  \npb{309}{1988}{259}. 

\bibitem{Laenen} E. Laenen, J. Smith, and W.L. van Neerven,
  \npb{369}{1992}{543}; \plb{321}{1994}{254}.

\bibitem{Berger} E.L. Berger and H. Contopanagos,
\plb{361}{1995}{115}.

\bibitem{Berends} F.A. Berends, J.B. Tausk, and W.T. Giele,
Phys. Rev. {\bf D47} (1993) 2746.

\bibitem{guesk1} S. G\"usken, J. H. K\"uhn and P. M. Zerwas,
                    \plb{155}{1985}{185}.
\bibitem{fadin1} V. Fadin, V. Khoze, \jl{46}{1987}{417};
                   \sovnp{48}{1988}{487}; \\
                   V. Fadin, V. Khoze and T. Sj\"ostrand,
 \zpc{48}{1990}{467}.
\bibitem{holli2} W. Hollik et al.~in Ref.\cite{reya1};\\
              W. Beenakker, A. Denner, W. Hollik et al., R. Mertig, T. Sack,
 and D. Wackeroth, \npb{411}{1994}{343}.

\bibitem{Lampe} B. Lampe, \plb{348}{1995}{196}.

\bibitem{Barger} V. Barger et al., \prd{35}{1987}{3366}.

\bibitem{KM} J.H. K\"uhn and E. Mirkes, \plb{311}{1993}{458}.

\bibitem{KP} J.H. K\"uhn and M. Peter, \zpc{67}{1995}{271}.

\bibitem{Letter} J.H. K\"uhn and E. Mirkes, \plb{296}{1992}{425}.
\bibitem{Paper} J.H. K\"uhn and E. Mirkes, \prd{48}{1993}{179}.


\bibitem{lighthiggs} A. Djouadi, M. Spira, and P.M. Zerwas,
  \plb{264}{1991}{440}. 

% part3 

\bibitem{fusi} O.J.P.~Eboli, G.C.~Marques, S.F.~Novales and
                A.A.~Natale, \prd{34}{1986}{771},\\
               S.~Dawson and S.~Willenbrock, \npb{284}{1987}{449},\\
               C.P.~Yuan, \npb{310}{1988}{1},\\
               R.P.~Kauffman, \prd{41}{1990}{3343},\\
                H.~Veltman, \prd{43}{1991}{2236}.
\bibitem{hagi}
       K.~Hagiwara, H.~Murayama and T.~Watanabe, \npb{367}{1991}{257}.
\bibitem{djo} A.~Djouadi, J.~Kalinowski and P.M.~Zerwas,
                \zpc{54}{1992}{255}.

\bibitem{DESY} J.H. K\"uhn, E. Mirkes, and J. Steegborn,
\zpc{57}{1993}{615}.. 

\bibitem{con} M.~Consoli, W.~Hollik and F.~Jegerlehner, in ``$Z$ Physics
              at LEP~1'', ed. G.~Altarelli, CERN89-08.
\bibitem{exac}
          J.~Jersak, E.~Laermann and P.M.~Zerwas,
          \prd{25}{1982}{363},\\
J.~Schwinger, ``Particles, Sources and Fields'', Addison-Wesley (1973),\\
         L.~Reinders, H.~Rubinstein and S.~Yazaki, \prc{127}{1985}{1}.

\bibitem{KueZer} J.H. K\"uhn and P.M. Zerwas, in {\em Advanced Series
    on Directions in HEP}, Vol.~10,  {\em Heavy Flavours},
  Eds. A.J. Buras and M. Lindner, World Scientific, 1992, p.~434.

\bibitem{appr} J.~Schwinger in \cite{exac}, and ref.~\cite{guesk1}.

\bibitem{HKT} A.H. Hoang, J.H. K\"uhn, and T. Teubner,
\npb{452}{1995}{173}. 

\bibitem{CKS} K. Chetyrkin, J.H. K\"uhn, and M. Steinhauser,
\plb{371}{1996}{93}. 

\bibitem{jez} M.~Je\.zabek, \zpc{56}{1992}{285}.
\bibitem{jad} S.~Jadach and B.F.L.~Ward, \cpc{56}{1990}{351}.
\bibitem{been1} W.~Beenakker, S.C.~van~der~Marck
              and W.~Hollik, \npb{365}{1991}{24}.

\bibitem{JKsim}
M. Je\.zabek and J.H. K\"uhn, Phys. Lett. {\bf B316} (1993) 360.
\bibitem{BH} W. Beenakker and W. Hollik, \plb{269}{1991}{425}.

\bibitem{GKKS}
B. Grzadkowski, P. Krawczyk, J.H. K\"uhn, and R.G. Stuart, Nucl. Phys. 
{\bf B281} (1987) 18.
\bibitem{fr1} C.~Peterson, D.~Schlatter, I.~Schmitt and P.M.~Zerwas,
               \pr{27}{83}{105}.
\bibitem{fr3} T.~Sj\"ostrand and P.M.~Zerwas, in Proceedings, {\em $e^+e^-$
       Collisions at 500 GeV: The Physics Potential}, DESY 92-123,
       pg.~463-475. 

\bibitem{fr2} G.~Jikia, \plb{257}{1991}{196}.
\bibitem{KOS} V.A. Khoze, L.H. Orr, and J. Sterling,
\npb{378}{1992}{413}. 

\bibitem{st1} G.L.~Kane, G.A.~Ladinsky and C.--P.~Yuan,
              \prd{45}{1992}{124}.\\
              C.--P.~Yuan, \prd{45}{1992}{782}.
\bibitem{BerOve} W. Bernreuther and P. Overmann, Proc. of the Workshop
  $e^+e^-$ Collisions at 500 GeV, the Physics Potential, Part C,
  P.M. Zerwas (ed.), DESY 93-123C, p. 287, and refs.~therein.

\bibitem{st2} C.~Schmidt and M.E.~Peskin, \prl{69}{1992}{410}.


\bibitem{kuehn2} J.H. K\"uhn, A. Reiter and P.M. Zerwas,
                   \npb{272}{1986}{560}.

\bibitem{st4} A.~Devoto, J.~Pumplin, W.~Repko and G.L.~Kane,
            \prl{43}{79}{1062}; 

\bibitem{jk41} J.H. K\"uhn, \npb{237}{1984}{77}.

%\bibitem{JKunp}
%      M.~Je\.zabek and J.H.~K\"uhn, Karlsruhe Univ. preprint TTP92-10.

\bibitem{Landau}
L. Landau and E. Lifshitz, {\em Non-relativistic Quantum Mechanics}, sect.~36.
\bibitem{Messiah}
A. Messiah, {\em Quantum Mechanics}, Chapter XI, sect.~7-11 
and Appendix B, sect.~4.

\bibitem{fk}
V.S.~Fadin and V.A.~Khoze, \jet{46}{1987}{525},


\bibitem{apple}
T.~Appelquist, M.~Dine and I.~Muzinich, \prd{17}{1978}{2074}.
\bibitem{sus}
L.~Susskind, in ``Weak and Electromagnetic Interactions at High
Energies'', Les Houches (1977). 
\bibitem{fis}
W.~Fischler, \npb{129}{1977}{157}.
\bibitem{bil}
A.~Billoire, \plb{92}{1980}{343}.
\bibitem{bern}
W.~Bernreuther and W.~Wetzel, \npb{197}{1982}{228}.
\bibitem{marc}
W.J.~Marciano, \prd{29}{1984}{580}.
\bibitem{JKT}  M.~Je\.zabek, J.H.~K\"uhn and T.~Teubner,
   \zpc{56}{1992}{653}

\bibitem{pesk}
J.M.~Strassler and M.E.~Peskin, \prd{43}{1991}{1500}.
\bibitem{TJ} M.~Je\.zabek and T.~Teubner, \zpc{59}{1993}{669}
\bibitem{Buchm}
For a pedagogical discussion see e.g.~the 
appendix of W. Buchm\"uller and S.-H. H. Tye, Phys. Rev. {\bf D24} (1981) 132.
\bibitem{Igo}
P. Igo-Kemenes, M. Martinez, R. Miguel, and S. Orteu,  in Proceedings,
{\em \epem\ Collisions 
at 500 GeV: The Physics Potential}, Part C; ed. P.M. Zerwas, 
DESY 93-123C, pg. 319.
\bibitem{GuthK}
R. Guth and J.H. K\"uhn,             Nucl. Phys. 
{\bf B368} (1992) 38.

\bibitem{JKm}
M. Je\.zabek and J.H. K\"uhn,  Phys. Lett. {\bf B207} (1988) 91.


\bibitem{Sumino} Y.~Sumino, K.~Fujii, K.~Hagiwara, H.~Murayama, C.-K.~Ng,
   \prd{47}{1992}{56}

\bibitem{JKTZer} M.~Je\.zabek, J.H.~K\"uhn and T.~Teubner,
 in Proceedings, {\em $e^+e^-$
       Collisions at 500 GeV: The Physics Potential}, DESY 92-123C,
pg.~303. 

\bibitem{peskin}  J.M.~Strassler and M.E.~Peskin, \prd{43}{1991}{1500}
\bibitem{MurSum2}
    H. Murayama and Y. Sumino, \prd{47}{1993}{82}.

\bibitem{hjkt}
    R. Harlander, M. Je\.zabek, J.H. K\"uhn and T. Teubner,
    \plb{346}{1995}{137}.

\bibitem{krz}
    J.H. K\"uhn, A. Reiter and P.M. Zerwas, \npb{272}{1986}{560}.
\bibitem{dev}
    A. Devoto, J. Pumplin, W. Repko and G.L. Kane,
    \prl{43}{1979}{1062}.
\bibitem{FKotsky}
    V.S. Fadin, V.A. Khoze and M.I. Kotsky, \zpc{64}{1994}{45}.

\bibitem{ttp48}
R. Harlander, M. Je\.zabek, J.H. K\"uhn, and M. Peter, Karlsruhe
preprint TTP 95-48, hep-ph/9604328, in print in Z.~Phys. {\bf C}.
\bibitem{MelYak1} K. Melnikov and O. Yakovlev, \plb{324}{1994}{217}.
\bibitem{sumph} Y. Sumino, PhD thesis, University of Tokyo, 1993
(unpublished). 
\bibitem{Huff} R.W.  Huff, {\it Ann. Phys.} {\bf 16} (1961) 288.
\bibitem{FMY94}
    K. Fujii, T. Matsui and Y. Sumino, \prd{50}{1994}{4341}.

\bibitem{Uber} H. \"Uberall, {\it Phys. Rev.} {\bf 119} (1960) 365. 
\bibitem{MKum} W. M\"odritsch and W. Kummer, \npb{430}{1994}{3}.


\end{thebibliography}

\addcontentsline{toc}{chapter}{Introduction}
\chapter*{Introduction}
An extensive search for top quarks has been performed at
electron-positron and hadron colliders for more than a decade.  First
evidence for top quark production in proton-antiproton collisions has
been announced by the CDF collaboration in the spring of 1994.  After
collecting  more luminosity subsequently both the CDF and the D0
experiments presented the definite analysis \cite{Sinerva}
 which demonstrated not only the existence of the
anti\-cipated quark but at the same time also provided a kinematic
determination of the top quark mass around 180 GeV and a production
cross section consistent with the QCD predictions.  The mass value is
in perfect agreement with the indirect mass determinations based on
precision measurements 
%\cite{bardin,HollikBrus,Swartz,Kniehl,Sirlin,degrassi} 
[2-7]
of the
electroweak 
parameters in \ep annihilation and in lepton-nucleon scattering.
Exploiting the quadratic top mass dependence of radiative corrections
an indirect mass measurement of 180 GeV with a present uncertainty of
roughly $20$  GeV  has been achieved.  

The top quark completes the fermionic spectrum of the Standard Model.
 All its properties are uniquely fixed after the mass has been
determined.  However, as a consequence of its large mass and decay
rate it will behave markedly different compared to the remaining five
lighter quarks.  

It is not just the obvious aim for completion which raises the
interest in the top quark.  With its mass comparable to the scale of
the electroweak symmetry breaking it is plausible that top quark
studies could provide  an important clue for the understanding of
fermion mass generation and the pattern of Yukawa couplings.  In fact,
it has been suggested that a top quark condensate could even be
responsible for the mechanism of spontaneous symmetry breaking
\cite{Hill}. 

These lectures will be mainly concerned with top quark phenomenology
within the framework of the Standard Model (SM).  
The precise understanding of its
production and decay constitutes the basis of any search for
deviations or physics beyond the SM. 

The properties of the top quark will be covered in chapter 1.  Direct
and indirect determinations of its decay rates, decay distributions
including QCD and electroweak corrections and decay modes predicted in
supersymmetric extensions will be discussed.  Top quark production at
hadron colliders will be the subject of chapter 2.  The production
cross section and momentum distributions are important ingredients in
any of the present analysis.   An alternative reaction, namely top
quark production through $W$-$b$-fusion allows to determine the
$W$-$b$-$\bar  t$ coupling and thus indirectly the top quark decay rate.

Perspectives for top studies at a future \ep collider will be
presented in chapter 3.  An accurate determination of the top quark
mass and its width to better than 1 GeV with a relative accuracy of
about 10\% seems feasible, and the electroweak couplings of the top
quark could be precisely measured with the help of polarized beams.
Of particular interest is the interplay between the large top quark
decay rate and the binding through the QCD potential which will be
also covered in chapter 3.
\setcounter{chapter}{0}